Author's response to reviews

Title: Weight and height z-scores improve after initiating ART among HIV-infected children in rural Zambia: a cohort study

Authors:

Catherine G Sutcliffe (csutclif@jhsph.edu)
Janneke H van Dijk (janneke.vandijk@miam.org.zm)
Bornface Munsanje (bornface.munsanje@miam.org.zm)
Francis Hamangaba (francis.hamangaba@miam.org.zm)
Pamela Sinywimaanzi (Pamela.sinywimaanzi@miam.org.zm)
Phil E Thuma (phil.thuma@macharesearch.org)
William J Moss (wmoss@jhsph.edu)

Version: 3 Date: 23 February 2011

Author's response to reviews:

Dear Editor,

we have made the requested changes:
1. we indicated that all authors have read and approved the final manuscript
2. we used initials when referring to the authors
3. we removed the tracking from the manuscript
4. we corrected any errors in the text
5. we put the abbreviations in a paragraph.